Author’s response to reviews

Title: A yoga intervention for diabetes risk reduction: A pilot randomized controlled trial

Authors:

Kelly A McDermott (mcdermottk@ocim.ucsf.edu)
M. Raghavendra Rao (raghay.hcgrf@gmail.com)
Raghuram Nagarathna (rnagaratna@gmail.com)
Elizabeth J Murphy (emurphy@medsfgh.ucsf.edu)
Adam Burke (aburke@sfsu.edu)
R. Hongasandra Nagendra (hrn@vyasa.org)
Fredrick M Hecht (rhecht@php.ucsf.edu)

Version: 6  Date: 6 June 2014

Author’s response to reviews: see over
June 5, 2014

RE: Revisions on MS: 7186885311091630 -- A yoga intervention for diabetes risk reduction: A pilot randomized controlled trial

Dear BMC CAM Editors:

We have updated the above referenced manuscript with more recent literature in response to the reviewers comment:

Since the current study is ~10 years old and there are several important limitations in this study, if authors had conducted any further studies on similar topic within this period, it would be highly recommended to include more meaningful data or provide reference in this publication...

And, in response to the Section editor’s comment:

I am fine with accepting this and just recommend that the authors make sure that the literature review is completely current.

We specifically included reference to a study by our coauthors in India that was concluded after the current pilot study. We have included a number of other references that discuss more recent findings for yoga in populations with both type 2 diabetes and at high risk of type 2 diabetes.

In addition, we have also provided a CONSORT checklist that refers to the page numbers of each section.

Sincerely,

Kelly McDermott, PhD  
Osher Center for Integrative Medicine  
University of California, San Francisco  
mcdermottk@ocim.ucsf.edu  
(206) 366-5271 (m)